Author's response to reviews

Title: Exploring the first delay: a qualitative study of home deliveries in Makwanpur district Nepal

Authors:

   Joanna Morrison (joanna.morrison@ucl.ac.uk)
   Rita Thapa (kunsang_rita@yahoo.com)
   Machhindra Basnet (pranms_life4@hotmail.com)
   Bharat Budhathoki (bbmagar3@gmail.com)
   Kirti Tumbahangphee (jetha_subba@hotmail.com)
   Dharma Manandhar (dsm@mira.org.np)
   Anthony Costello (anthony.costello@ucl.ac.uk)
   Osrin David (d.osrin@ucl.ac.uk)

Version: 2 Date: 7 August 2013

Author's response to reviews: see over
7th August 2013

Dear Janelyn Cruz,

Exploring the first delay: a qualitative study of home deliveries in Makwanpur district Nepal

We apologise for submitting an incomplete manuscript. We have now uploaded the correct manuscript with:
1. A title page providing full names, email addresses and institutional addresses for all authors, and indicating the corresponding author.
2. An abstract with the required sections that is 347 words, with key words added
3. A competing interests statement
4. Details of author contributions

Thank you for your consideration
Yours sincerely

Joanna Morrison PhD
Senior Research Associate
University College London